Author’s response to reviews

Title: The Impact of IgM Deposits on the Outcome of Nephrotic Syndrome in Children

Authors:
Sandra Juozapaite (sandra.melnik10@gmail.com)
Rimante Cerkauskiene (rimante.cerkauskiene@vuvl.lt)
Arvydas Laurinavicius (Arvydas.Laurinavicius@vpc.lt)
Augustina Jankauskiene (augustinajankauskiene@yahoo.com)

Version: 4 Date: 01 Jun 2017

Author’s response to reviews:

1. Please remove any figure titles and captions from the figure files as these files should contain the image graphics only. Upon doing this, please place the titles/captions at the end of the main manuscript after the References under the heading “Figure Legend”.

Corrected.

2. Thank you for providing the authorship form. However, please provide more justification for the contributions of authors. Currently, the contributions of RC and AJ do not automatically qualify for authorship. Please amend your manuscript accordingly.

Corrected.

3. Please include a statement on consent to participate in the “Ethics approval and consent to participate” section of the Declarations.

Ethics approval was confirmed by Lithuanian Bioethics’ Committee Manifest No 42 in concert with Children’s Hospital, affiliate of Vilnius University Hospital Santariskiu Clinics Ethics Committee consent. The consent from the patients was not requested since this study was retrospective and did not include active interaction with the patients.

4. Please submit your revised manuscript as a clean copy without any tracked changes, coloured or highlighted text, as these are no longer required at this stage of the editorial process.

Corrected.